# The ODYSSEY DM-DYSLIPIDEMIA trial: confirming the benefits of alirocumab in diabetic mixed dyslipidemia

## Paul Chan<sup>1\*</sup>, Li Shao<sup>2\*</sup>, Brian Tomlinson<sup>3,4</sup>, Zhong-Min Liu<sup>5</sup>

<sup>1</sup>Division of Cardiology, Department of Internal Medicine, Wan Fang Hospital, Taipei Medical University, Taipei, Taiwan; <sup>2</sup>The VIP Department, Shanghai East Hospital, Tongji University School of Medicine, Shanghai 200120, China; <sup>3</sup>Research Center for Translational Medicine, Shanghai East Hospital Affiliated to Tongji University School of Medicine, Shanghai 200120, China; <sup>4</sup>Department of Medicine & Therapeutics, Chinese University of Hong Kong, Shatin, Hong Kong, China; <sup>5</sup>Department of Cardiac Surgery, Shanghai East Hospital, Tongji University, Shanghai 200120, China \*These authors contributed equally to this work.

*Correspondence to:* Brian Tomlinson. Research Center for Translational Medicine, Shanghai East Hospital Affiliated to Tongji University School of Medicine, Shanghai 200120, China and Department of Medicine & Therapeutics, Chinese University of Hong Kong, Shatin, Hong Kong, China. Email: btomlinson@cuhk.edu.hk; Zhong-Min Liu. Department of Cardiac Surgery, Shanghai East Hospital, Tongji University, Shanghai 200120, China. Email: liu.zhongmin@tongji.edu.cn.

*Provenance:* This is an invited Editorial commissioned by Section Editor Dr. Kaiping Zhang, PhD (AME College, AME Group, Hangzhou, China). *Comment on:* Müller-Wieland D, Leiter LA, Cariou B, *et al.* Design and rationale of the ODYSSEY DM-DYSLIPIDEMIA trial: lipid-lowering efficacy and safety of alirocumab in individuals with type 2 diabetes and mixed dyslipidaemia at high cardiovascular risk. Cardiovasc Diabetol 2017;16:70.

Submitted Oct 19, 2017. Accepted for publication Oct 26, 2017. doi: 10.21037/atm.2017.10.26 **View this article at:** http://dx.doi.org/10.21037/atm.2017.10.26

The role of triglycerides in the development of atherosclerotic cardiovascular disease (ASCVD) has been a controversial issue for many years, but recent evidence from epidemiology and Mendelian randomization studies has confirmed that elevated triglycerides, and perhaps more importantly triglyceride-rich lipoproteins, are strong and independent predictors of ASCVD and allcause mortality (1). Hypertriglyceridemia is often related to genetic factors which are typically polygenic in mildto-moderate hypertriglyceridemia or monogenic in severe hypertriglyceridemia with triglyceride concentration >10 mmol/L (>885 mg/dL) (2). Lifestyle, other diseases and medications also influence triglyceride levels and diabetes is typically associated with elevated triglycerides, reduced high-density lipoprotein cholesterol (HDL-C) and lowdensity lipoprotein (LDL) particles which are small and dense resulting in atherogenic diabetic dyslipidemia (3).

The contribution of cholesterol carried in triglyceriderich lipoproteins in addition to LDL cholesterol (LDL-C) can be estimated by calculating the non-HDL-C which should then reflect the total burden of cholesterol transported in atherogenic lipoproteins. Non-HDL-C levels provide better prediction of ASCVD events than LDL-C levels in statin-treated patients (4) and in an analysis of changes in atheroma volume determined by coronary intravascular ultrasonography, the achieved non-HDL-C levels were more closely associated with coronary atheroma progression than LDL-C levels (5). Based on the current understanding, some international lipid management guidelines recommend non-HDL-C levels as alternative primary or secondary targets of treatment in addition to LDL-C levels in all patients (6,7) or specifically in patients with diabetes (8).

The ODYSSEY DM-DYSLIPIDEMIA trial was designed to examine the effects of alirocumab in an important group of patients with type 2 diabetes mellitus (T2DM) with mixed dyslipidemia at high cardiovascular risk, a group that has not previously been studied prospectively in the development programs with the proprotein convertase subtilisin-kexin type 9 (PCSK9) monoclonal antibodies (mAbs) (9). Likewise, the other ODYSSEY trial in diabetic patients, the ODYSSEY DM-INSULIN trial, focused on another novel group of patients with either type 1 diabetes mellitus (T1DM) or T2DM who were treated with insulin, again a group that has not been studied prospectively before (10). Both trials have now been completed and the results were presented at the meetings of the American Diabetes Association (ADA) in San Diego and the European Association for the Study of Diabetes (EASD) in Lisbon in 2017. The ODYSSEY DM-INSULIN trial has been published recently (11).

Other trials in the ODYSSEY program for alirocumab (12) and the PROFICIO clinical trial program with evolocumab (13) have focused on the reduction in LDL-C in patients with primary hypercholesterolemia, either familial or non-familial. The inclusion criteria usually included a baseline LDL-C above the target according to the level of cardiovascular risk, such as LDL-C  $\geq$ 1.8 mmol/L (70 mg/dL) for very highrisk patients. Sometimes a target for non-HDL-C which was 0.8 mmol/L (30 mg/dL) above the LDL-C target was an alternative lipid inclusion criterion. This difference in values between LDL-C and non-HDL-C targets is based on the threshold for elevated fasting triglycerides being >1.7 mmol/L (>150 mg/dL) which converts into these figures for cholesterol carried in triglyceride-rich lipoproteins using the Friedewald formula. These trials did not exclude patients with T2DM but generally excluded patients with baseline fasting triglyceride levels >4.5 mmol/L (>400 mg/dL), partly because calculated baseline LDL-C levels become inaccurate with levels of triglyceride about this value.

Considering that the PCSK9 mAbs have a primary effect to reduce circulating PCSK9 and thus reduce LDL-C by upregulating the LDL receptors (LDLRs) (14), the primary endpoint of most of the studies was the change in LDL-C, which was assessed after 24 weeks with alirocumab or after 12 weeks with evolocumab (12,13). The studies also reported significant decreases in other atherogenic lipid parameters including total cholesterol, non-HDL-C, apolipoprotein (apo) B, triglycerides and lipoprotein(a) [Lp(a)], and there were usually increases in HDL-C and apoAI (15). The average relative reductions in LDL-C with alirocumab given in subcutaneous doses of 75 or 150 mg every 2 weeks have been fairly consistent at about 40% to 60% in most patient groups, irrespective of baseline values, although there is a considerable variation in response between individuals. Patients with homozygous familial hypercholesterolemia usually respond less well and those with null mutations in the LDLR gene did not respond at all in the studies with evolocumab (16). Patients on statin treatment show some differences in the pharmacokinetics but not the pharmacodynamics with alirocumab and

evolocumab (12,13). However, there is no a priori reason to expect that any other group of patients would show a diminished response to PCSK9 mAbs.

A recent analysis evaluated the efficacy of alirocumab on non-HDL-C and apoB from data derived from 4,983 patients enrolled in ten randomized, placebo- or ezetimibecontrolled phase 3 ODYSSEY trials. The primary end point for this pooled analysis was percent reduction in non-HDL-C and apoB at week 24 (17). For the trials in which alirocumab was compared to placebo in the ODYSSEY COMBO I, FH I, and FH II studies, the alirocumab starting dose was 75 mg every 2 weeks (Q2W) and it was then increased to 150 mg Q2W at week 12 if the predefined ASCVD risk-based LDL-C goals were not achieved at week 8. In the ODYSSEY LONG TERM and HIGH FH trials, the starting dose of alirocumab was 150 mg Q2W. In these two groups of placebo-controlled ODYSSEY trials, the least squares mean differences [95% confidence intervals (CI)] compared to placebo were, respectively, -46.4% (-49.6% to -43.2%) and -51.6% (-53.7% to -49.5%) for non-HDL-C, -41.3% (-44.0% to -38.6%) and -52.9% (-55.2% to -50.7%) for apoB, -10.3% (-14.0% to -6.6%) and -17.0% (-19.7% to -14.3%) for triglyceride, -52.7% (-56.3% to -49.2%) and -60.9% (-63.3% to -58.5%) for LDL-C and 7.6% (5.6% to 9.6%) and 4.5% (3.3% to 5.8%) for HDL-C (17).

This analysis and other trials with the PCSK9 mAbs have generally shown relative reductions in triglycerides that are substantially less than the reductions in LDL-C, a situation similar to the effects of statins (18). Mechanistic studies in healthy subjects showed that alirocumab increased the clearance of intermediate-density lipoprotein (IDL) as well as LDL and there was also an increase in the fractional clearance rate of Lp(a) with no change in the production rate, suggesting increased LDLRs may play a role in the reduction of plasma Lp(a) (19). A study with evolocumab showed similar findings but also showed increased fractional catabolism of very-low-density lipoprotein (VLDL)-apoB and reduced production rate of IDL-apoB suggesting that the clearance of VLDL particles from the circulation may also be increased (20). PCSK9 also interacts with other members of the LDLR superfamily including the VLDL receptor (VLDLR), apolipoprotein E receptor 2 (ApoER2), and lipoprotein receptor-related protein 1 (LRP1), and effects of PCSK9 mAbs to reduce the PCSK9 interaction with these receptors may contribute to increased triglyceride-rich lipoprotein catabolism, and thereby reduction of triglyceride levels, as well as the effects on the

#### LDLR (21).

There are no previous trials with PCSK9 mAbs in patients selected for mixed hyperlipidemia but an analysis was performed of the effects of evolocumab in 1,148 patients selected from four 12-week phase 3 randomized studies which compared those with mixed hyperlipidemia who had fasting triglyceride levels  $\geq 1.7$  mmol/L  $(\geq 150 \text{ mg/dL})$  to those with triglycerides <1.7 mmol/L (22). The mean of the change from baseline in LDL-C at weeks 10 and 12 with evolocumab was approximately -67% versus placebo and -42% versus ezetimibe (all P<0.001) compared to -65% versus placebo and -39% versus ezetimibe in participants with and without elevated triglycerides, respectively. There was a similar pattern of treatment differences for evolocumab versus placebo and ezetimibe for non-HDL-C and apoB so that overall the reductions of atherogenic lipids with evolocumab were similar in patients with and without mixed hyperlipidemia.

Likewise, no previous trials have only enrolled patients with T2DM but a meta-analysis based on individual participant data of the clinical trials with evolocumab in adult patients with or without T2DM, excluding trials that included patients with homozygous familial hypercholesterolemia, identified three trials that included 413 patients with T2DM and 2,119 patients without T2DM (23). The mean reductions in LDL-C were 60% (95% CI, 51-69%) versus placebo in patients with T2DM and 66% (95% CI, 62-70%) versus placebo in those without T2DM. Reductions in non-HDL-C were 55% (95% CI, 47-63%) versus placebo in patients with T2DM and 58% (95% CI, 55-61%) in those without T2DM and reductions in triglycerides were 23% (95% CI, 12-34%) in patients with T2DM compared with 17% (95% CI, 14-21%) in those without T2DM. Findings were similar across diabetes subgroups based on glycemia, insulin use, renal function, and cardiovascular disease status so overall the lipid responses were not different between patients with and without T2DM.

The FOURIER (Further Cardiovascular Outcomes Research with PCSK9) trial with evolocumab included 11,031 patients (40%) who had diabetes and 16,533 (60%) who did not have diabetes at baseline and a prespecified analysis investigated the effect of evolocumab on cardiovascular events by diabetes status (24). Evolocumab reduced cardiovascular outcomes significantly and to the same extent in patients with and without diabetes at baseline. The hazard ratios (HRs) were 0.83 (95% CI, 0.75–0.93; P=0.0008) and 0.87 (0.79–0.96; P=0.0052) for patients with and without diabetes (P interaction =0.60), respectively, for the primary composite endpoint of cardiovascular death, myocardial infarction, stroke, hospital admission for unstable angina, or coronary revascularization. The HRs for the key secondary endpoint, which was a composite of cardiovascular death, myocardial infarction, or stroke, were also similar in patients with and without diabetes (P interaction =0.65). There was no increased risk of newonset diabetes with evolocumab in patients without diabetes at baseline, including in those with prediabetes and levels of glycated hemoglobin (HbA1c) and fasting plasma glucose (FPG) were similar between the evolocumab and placebo groups over time in all patients grouped according to baseline glycemic status.

In the ODYSSEY DM-INSULIN study, there were 441 participants with T2DM and 76 with T1DM with baseline LDL-C  $\geq$ 70 mg/dL ( $\geq$ 1.8 mmol/L) and increased cardiovascular risk (11). They were randomized to receive subcutaneous alirocumab or placebo in a 2:1 ration Q2W for 24 weeks. The alirocumab dose was initially 75 mg Q2W and this was increased to 150 mg Q2W at week 12 if necessary, as in other alirocumab studies. The mean (± standard error) reductions in LDL-C from baseline to week 24 were 49.0%±2.7% and 47.8%±6.5% versus placebo (both P<0.0001) in patients with T2DM and T1DM, respectively. There were also significant reductions in non-HDL-C, apoB, triglycerides and Lp(a). The non-HDL-C target of <100 mg/dL (<2.6 mmol/L) was achieved in 79% and 71% of T2DM and T1DM patients, respectively. Most of the T2DM (80%) and T1DM (63%) patients did not need the dose of alirocumab to be increased and remained on the starting dose of 75 mg Q2W. There was no significant change in HbA1c and FPG levels and concomitant administration of alirocumab and insulin did not raise any safety concerns.

Taken together, the available data indicate that the lipid changes with alirocumab would be similar in patients with and without mixed hyperlipidemia and in patients with and without T2DM. Moreover, the cardiovascular outcome with alirocumab might be expected to be similar in patients with and without diabetes, based on the FOURIER cardiovascular outcome trial with evolocumab. There is also no evidence to date that the PCSK9 mAbs increase the risk of development of new onset diabetes or worsen glycemic control in patients with established diabetes or impaired glucose tolerance.

The design of the ODYSSEY DM-DYSLIPIDEMIA study was to enroll 420 individuals with T2DM with

#### Page 4 of 5

documented ASCVD or  $\geq 1$  additional cardiovascular risk factor and mixed dyslipidemia, defined as non-HDL-C  $\geq$ 2.59 mmol/L (100 mg/dL) and triglycerides  $\geq$ 1.7 and <5.65 mmol/L (≥150 and <500 mg/dL), with non-HDL-C inadequately controlled despite maximally tolerated statin therapy (25). Participants were randomized (2:1) to additional treatment with alirocumab 75 mg every 2 weeks (Q2W) with a blinded dose increase to 150 mg Q2W at week 12 if non-HDL-C ≥100 mg/dL at week 8, or to lipidlowering usual care. The usual care allowed the option of adding the pre-specified choice of ezetimibe, fenofibrate, omega-3 fatty acids or nicotinic acid, in accordance with local standard-of-care. In the 413 patients recruited from 14 countries worldwide, the primary efficacy endpoint reported was a reduction in non-HDL-C of 33.3% with alirocumab compared with fenofibrate (P<0.0001).

In conclusion, the ODYSSEY DM-DYSLIPIDEMIA trial fills an important gap in the development programs of the PCSK9 mAbs. Patients with T2DM are a group with high ASCVD risk and typically have mixed dyslipidemia. It may be relatively easy to achieve target LDL-C levels in these patients but it is important to consider non-HDL-C targets, which are less often attained with current therapies. The results presented from the ODYSSEY DM-DYSLIPIDEMIA trial with alirocumab are encouraging and considering the data available from the analysis of completed trials with alirocumab and evolocumab, the findings were not surprising. We await the final publication of the trial with interest.

#### Acknowledgements

None.

### Footnote

*Conflicts of Interest*: B Tomlinson has received research funding from Amgen, AstraZeneca, Merck Serono, Merck Sharp and Dohme, Novartis, Pfizer and Roche, and he has acted as consultant, advisor or speaker for Amgen, AstraZeneca, Merck Serono, Merck Sharp and Dohme and Sanofi. The other authors have no conflicts of interest to declare.

#### References

1. Nordestgaard BG. Triglyceride-Rich Lipoproteins and Atherosclerotic Cardiovascular Disease: New Insights

From Epidemiology, Genetics, and Biology. Circ Res 2016;118:547-63.

- Hegele RA, Ginsberg HN, Chapman MJ, et al. The polygenic nature of hypertriglyceridaemia: implications for definition, diagnosis, and management. Lancet Diabetes Endocrinol 2014;2:655-66.
- Adiels M, Olofsson SO, Taskinen MR, et al. Diabetic dyslipidaemia. Curr Opin Lipidol 2006;17:238-46.
- Boekholdt SM, Arsenault BJ, Mora S, et al. Association of LDL cholesterol, non-HDL cholesterol, and apolipoprotein B levels with risk of cardiovascular events among patients treated with statins: a meta-analysis. JAMA 2012;307:1302-9.
- Puri R, Nissen SE, Shao M, et al. Non-HDL Cholesterol and Triglycerides: Implications for Coronary Atheroma Progression and Clinical Events. Arterioscler Thromb Vasc Biol 2016;36:2220-8.
- 6. Catapano AL, Graham I, De Backer G, et al. 2016 ESC/ EAS Guidelines for the Management of Dyslipidaemias: The Task Force for the Management of Dyslipidaemias of the European Society of Cardiology (ESC) and European Atherosclerosis Society (EAS) Developed with the special contribution of the European Assocciation for Cardiovascular Prevention & Rehabilitation (EACPR). Eur Heart J 2016;37:2999-3058.
- Anderson TJ, Gregoire J, Pearson GJ, et al. 2016 Canadian Cardiovascular Society Guidelines for the Management of Dyslipidemia for the Prevention of Cardiovascular Disease in the Adult. Can J Cardiol 2016;32:1263-82.
- Lloyd-Jones DM, Morris PB, Ballantyne CM, et al. 2016 ACC Expert Consensus Decision Pathway on the Role of Non-Statin Therapies for LDL-Cholesterol Lowering in the Management of Atherosclerotic Cardiovascular Disease Risk: A Report of the American College of Cardiology Task Force on Clinical Expert Consensus Documents. J Am Coll Cardiol 2016;68:92-125.
- Müller-Wieland D, Leiter LA, Cariou B, et al. Design and rationale of the ODYSSEY DM-DYSLIPIDEMIA trial: lipid-lowering efficacy and safety of alirocumab in individuals with type 2 diabetes and mixed dyslipidaemia at high cardiovascular risk. Cardiovasc Diabetol 2017;16:70.
- Cariou B, Leiter LA, Muller-Wieland D, et al. Efficacy and safety of alirocumab in insulin-treated patients with type 1 or type 2 diabetes and high cardiovascular risk: Rationale and design of the ODYSSEY DM-INSULIN trial. Diabetes Metab 2017;43:453-9.
- 11. Leiter LA, Cariou B, Müller-Wieland D, et al. Efficacy and safety of alirocumab in insulin-treated individuals with

#### Annals of Translational Medicine, Vol 5, No 23 December 2017

- Tomlinson B, Hu M, Zhang Y, et al. Alirocumab for the treatment of hypercholesterolemia. Expert Opin Biol Ther 2017;17:633-43.
- Tomlinson B, Hu M, Zhang Y, et al. Evolocumab for the treatment of hypercholesterolemia. Expert Opin Biol Ther 2017;17:1447-61.
- Lambert G, Charlton F, Rye KA, et al. Molecular basis of PCSK9 function. Atherosclerosis 2009;203:1-7.
- Roth EM, Diller P. Alirocumab for hyperlipidemia: physiology of PCSK9 inhibition, pharmacodynamics and Phase I and II clinical trial results of a PCSK9 monoclonal antibody. Future Cardiol 2014;10:183-99.
- 16. Raal FJ, Hovingh GK, Blom D, et al. Long-term treatment with evolocumab added to conventional drug therapy, with or without apheresis, in patients with homozygous familial hypercholesterolaemia: an interim subset analysis of the open-label TAUSSIG study. Lancet Diabetes Endocrinol 2017;5:280-90.
- Bays HE, Leiter LA, Colhoun HM, et al. Alirocumab Treatment and Achievement of Non-High-Density Lipoprotein Cholesterol and Apolipoprotein B Goals in Patients With Hypercholesterolemia: Pooled Results From 10 Phase 3 ODYSSEY Trials. J Am Heart Assoc 2017;6. pii: e005639.
- Karlson BW, Palmer MK, Nicholls SJ, et al. A VOYAGER Meta-Analysis of the Impact of Statin Therapy on Low-Density Lipoprotein Cholesterol and Triglyceride Levels

**Cite this article as:** Chan P, Shao L, Tomlinson B, Liu ZM. The ODYSSEY DM-DYSLIPIDEMIA trial: confirming the benefits of alirocumab in diabetic mixed dyslipidemia. Ann Transl Med 2017;5(23):477. doi: 10.21037/atm.2017.10.26 in Patients With Hypertriglyceridemia. Am J Cardiol 2016;117:1444-8.

- Reyes-Soffer G, Pavlyha M, Ngai C, et al. Effects of PCSK9 Inhibition With Alirocumab on Lipoprotein Metabolism in Healthy Humans. Circulation 2017;135:352-62.
- Watts GF, Chan DC, Dent R, et al. Factorial Effects of Evolocumab and Atorvastatin on Lipoprotein Metabolism. Circulation 2017;135:338-51.
- Seidah NG, Awan Z, Chretien M, et al. PCSK9: a key modulator of cardiovascular health. Circ Res 2014;114:1022-36.
- 22. Rosenson RS, Jacobson TA, Preiss D, et al. Efficacy and Safety of the PCSK9 Inhibitor Evolocumab in Patients with Mixed Hyperlipidemia. Cardiovasc Drugs Ther 2016;30:305-13.
- 23. Sattar N, Preiss D, Robinson JG, et al. Lipid-lowering efficacy of the PCSK9 inhibitor evolocumab (AMG 145) in patients with type 2 diabetes: a meta-analysis of individual patient data. Lancet Diabetes Endocrinol 2016;4:403-10.
- Sabatine MS, Leiter LA, Wiviott SD, et al. Cardiovascular safety and efficacy of the PCSK9 inhibitor evolocumab in patients with and without diabetes and the effect of evolocumab on glycaemia and risk of new-onset diabetes: a prespecified analysis of the FOURIER randomised controlled trial. Lancet Diabetes Endocrinol 2017;5:941-50.
- Elfar A, Thompson PD. Variability in low density lipoprotein-cholesterol concentrations after alirocumab injection. J Clin Lipidol 2017;11:307.